Author's response to reviews

Title: Loss of neuromonitoring signal during bilateral thyroidectomy: does it change the operative strategy?

Authors:

Lilly Khamsy (lillykhamsy@hotmail.ch)
PauL E Constantin (Paul.Constanthin@hcuge.ch)
Samira M Sadowski (samirasadowski@hotmail.com)
Frederic Triponez (frederic.triponez@hcuge.ch)

Version: 6 Date: 28 July 2015

Author's response to reviews:

Dear Editor,

Please find here the latest version of our manuscript.

All the numbers are correct; as is usual in most studies, there are some missing data, which is also clearly mentioned in the manuscript (69 answers..and 66 (96%) of the questionnaires were fully completed.) The percentages are calculated according to the number of responses obtained, which is the reason why the same number does not exactly correspond to the same percentage in different Tables.

The 3 Tables are included in the main manuscript.

Best regards

FT